Author's response to reviews

Title: The relationship between obesity and neurocognitive function in Chinese patients with schizophrenia

Authors:

  xiaofeng Guo (gxfd@sina.com.cn)
  zhanchou Zhang (zzcd@163.com)
  Qinling Wei (qinlingwei001@163.com)
  Hailong Lv (bingquan1234@126.com)
  Renrong Wu (zhaijinguo002@126.com)
  Jingping Zhao (fengcsu@yahoo.com.cn)

Version: 5 Date: 18 March 2013

Author's response to reviews: see over
Response to Reviewer’s Comments

We thank the reviewers for their helpful comments and suggestions regarding the above-referenced manuscript. Please see below for our point-by-point response. For clarity we have directly quoted or paraphrased the reviewers’ specific concerns or critiques in italics, then provided our response.

I think that the authors have responded adequately for all the comments except for one minor essential comments:

The authors must describe in the abstract what was meant by the "composite z score".

Response 1:

Thank you for the good suggestion. We have added this description in the abstract.

Quality of written English: Needs some language corrections before being Published

Response 1:

Thank you for the good suggestion. We have invited Elizabeth W. Twamley from Department of Psychiatry, University of California to proof read the manuscript. We have also revised the entire manuscript according to the journal style.